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NuMI Update

• Down time Tuesday and Wednesday for tests of accelerator 
safety systems and maintenance 

• Target scan done Friday all parameters nominal, no water around 
it, and target condition looks good

• Ran at 12 turns in the booster with occasionally difficulties from 
beam loss



Beam Delivered Last Week
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Beam Delivered Since May 1, 2005

Daily Protons Delivered to NuMI Target Starting 1 May 2005
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About 9 x 1019 protons delivered in total 

Total NuMI Protons on Target Starting 1 May 
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MINOS Status

• Far Detector

• Uptime still near 100%

• 1 problem channel in the veto shield, but that does not affect beam 
data taking

• Near Detector

• Uptime at 99% for the week

• 3 bad channels swapped



Detector Performance



Target Longevity

• Plots show number of ND events/day normalized by POT to look for changes in 
the target performance

• Separate events into 2 energy bins as indicated - low energies points to problems 
with the horn, high energies can point to problems with the target


